Author's response to reviews

Title: Does lowering alcohol use among Dutch high school students lead to higher use of tobacco and cannabis?

Authors:

Claudia E Verhagen (cverhagen@ggd.amsterdam.nl)
Daan G Uitenbroek (duitenbroek@ggd.amsterdam.nl)
Emilie J Schreuders (cverhagen@ggd.amsterdam.nl)
Sabah El Messaoudi (cverhagen@ggd.amsterdam.nl)
Marlou LA de Kroon (m.dekroon@erasusmc.nl)

Version: 6  Date: 29 July 2015

Author's response to reviews:

Dear Sir, Madam,

Thank you very much for your further comments on our paper. We now cite the legal issue at the appropriate place and changed the numbering in the literature list accordingly. We hope that everything is fine now. However, do not hesitate to contact us again if there are further problems.

Sincerely,

Daan Uitenbroek